Author’s response to reviews

Title: Engaging participants in a complex intervention trial in Australian General Practice

Authors:

David Perkins (dperkins@gwahs.health.nsw.gov.au)
Mark F Harris (m.f.harris@unsw.edu.au)
Jocelyn Tan (j.tan@unsw.edu.au)
Bettina Christl (b.christl@unsw.edu.au)
Jane Taggart (j.taggart@unsw.edu.au)
Mahnaz Fanaian (m.fanaian@unsw.edu.au)

Version: 2 Date: 5 June 2008

Author’s response to reviews:

Dr Melissa Norton MD
Editor
BMC Family Practice

Dear Dr Norton

Thank you for re-considering our research paper “Engaging participants in a complex intervention trial in Australian General Practice” for publication in your journal.

We would like to thank the reviewers for their thoughtful comments. We have revised the paper along the lines suggested by the reviewers and we believe it is much improved as a result.

Yours sincerely,

Mark Harris
On Behalf of the authors
June 5 2008